Author's response to reviews

Title: Adherence to anti diabetic medication among patients with diabetes in eastern Uganda; a cross sectional study

Authors:

James Bagonza (jbagonza@musph.ac.ug)
Elizeus Rutebemberwa (ellie@musph.ac.ug)
William Bazeyo (wbazeyo@musph.ac.ug)

Version: 7 Date: 19 February 2015

Author's response to reviews:

Resubmission date: February 19, 2015
The Editor
BMC Health Services Research
Dear Editor,
Re: Resubmission of manuscript MS: 9384133211326114 titled Adherence to anti diabetic medication among patients with diabetes in eastern Uganda; a cross sectional study
We are pleased to resubmit our manuscript entitled: “Adherence to anti diabetic medication among patients with diabetes in eastern Uganda; a cross sectional study”. We thank the reviewers for the constructive and supportive comments. Below is a point by point response to the reviewers’ comments.
Point by point response to reviewers’ comments
Reviewer 1: Jayadevan Sreedharan
Reviewer’s report:
No more comments on this. The manuscript is ok
Level of interest: An article of importance in its field
Quality of written English: Needs some language corrections before being published
Response: We thank the reviewer for this comment. The manuscript has been reviewed by a native English speaker and the changes can be seen from the attached manuscript showing tracked changes.
Statistical review: Yes, and I have assessed the statistics in my report.
Reviewer 2: Mary Justin-Temu
Reviewer’s report:
On page 10 from the bottom second sentence which reads "Unfortunately in rural areas, few patients can afford such costs" The last word is too complex I suggest
to replace with the word costs" with "price". This was an over-site in the previous review.
Response: We thank the reviewer for this comment and the word “costs” has been replaced with “price” as can be seen on p. 10 fifth line from the bottom.
Space between the Subtopics and paragraphs will be OK.
Level of interest: An article of importance in its field
Quality of written English: Acceptable
Statistical review: No, the manuscript does not need to be seen by a statistician.

Once again, we thank you editors for the work you do for us.

Yours sincerely,

Dr. Elizeus Rutebemberwa
Corresponding author
Department of Health Policy Planning and Management
Makerere University School of Public Health, College of Health sciences,
Kampala Uganda
P O Box 7072 Kampala Uganda
Mulago Hospital Complex
Email: ellie@musph.ac.ug
Phone: +256 700 720077